Author's response to reviews

Title: Chylous ascites as the main manifestation of left ventricular dysfunction: a case report

Authors:
Ezequiel Ridruejo (eridruejo@cemic.edu.ar)
Oscar G Mando (omando@cemic.edu.ar)

Version: Date: 21 July 2005

Author's response to reviews:

Dear Elizabeth C. Moylan PhD
Assistant Editor
BioMed Central

I have changed the title to Chylous ascites as the main manifestation of left ventricular dysfunction: a case report.
I have removed underline from title
I have removed qualifications from the list of authors.
I have written each affiliation in full and have provided in the following format: department, institute, city, country. I do not use full capitals for country names, and I do not include full-stops in abbreviated country names.
I have linked each author to their corresponding affiliation with the use of superscript numbers.

I have removed the keywords section from the manuscript.
I have removed this from the manuscript.
I have used sentence case for all headings and sub-headings in the manuscript. I have placed the word abstract on a separate line above Background - rather than have them close together. I have checked all section headings.

Manuscript sections include (in the following order): Abstract; Background; Case presentation; Discussion; Conclusions; Competing interests; Authors' contributions; References; Tables.
INTRODUCTION was replaced by Background. CASE REPORT was replaced by presentation. As told by Elizabeth C. Moylan PhD, Assistant Editor, the discussion section was kept and a conclusions section added.
I have modified the references: should be cited in the text using consecutive numbers in square brackets i.e. [1]. At the top of page 6 - I you really mean ref 12-14. The reference list contains all authors' names, regardless of how many there are. The term "et al" was removed.
I have included a 'Competing interests' section between the Conclusions and Authors' contributions.

The answers to the questions that are asked of authors are:
Financial competing interests
In the past five years have * you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.
No.
Do you hold any stocks or * shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.
No.
Do you hold or are you * currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.
No.
Do you have any other * financial competing interests? If so, please specify.
No.
Non-financial competing interests
Are there any non-financial competing interests (political, personal, religious, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.
No.
An Authors' contributions section has been included.
Tables have been incorporated into the main manuscript.
Sincerely yours,
Ezequiel Ridruejo MD.
Hepatology Section, Department of Medicine. Centro de Educacion Medica e Investigaciones Clinicas
Norberto Quirno "CEMIC".
Email: eridruejo@cemic.edu.ar
Telephone: 54-11-4809-1980
Fax: 54-11-4809-1992
Avda. Las Heras 2939
(C1425ASG) Buenos Aires, Argentina